Author's Response To Reviewer Comments

Clo<u>s</u>e

1. I have included the corrected author list

 I citing the GigaDB repository. However, since the doi is not yet availabl, I followed this advise by Chris Hunter: "Thank you, I will process the dataset as soon as possible. If that is the only bit of information you require to complete the manuscript submission then there is no need for you to wait, the handling editor (CC'd) can insert the dataset citation on your behalf during the copy editing stage."
I send a brief explanation via email regarding the additional author.

4. The additional author fits under the contribution listed as was in place.

Clo<u>s</u>e